Journal of Cleaner Production 219 (2019) 833-845

Contents lists available at ScienceDirect

Journal of Cleaner Production

journal homepage: www.elsevier.com/locate/jclepro



Measuring ecological capital: State of the art, trends, and challenges



Huajun Yu^a, Yutao Wang^{a, b, *}, Xiao Li^c, Chengdong Wang^a, Mingxing Sun^d, Anshu Du^a ^a Fudan Tyndall Center and Shanghai Key Laboratory of Atmospheric Particle Pollution and Prevention (LAP³), Department of Environmental Science & Engineering, Fudan University, Shanghai, 200438, China

^b Institute of Eco-Chongming (SIEC), No. 3663 Northern Zhongshan Road, Shanghai, 200062

^c Center for Industrial Ecology, School of Forestry and Environmental Studies, Yale University, 195 Prospect St, New Haven, CT, 06511, USA

^d International Ecosystem Management Partnership, Institute of Geographic Sciences and Natural Resources Research, Chinese Academy of Sciences, Beijing,

100101, PR China

ARTICLE INFO

Article history: Received 31 October 2018 Received in revised form 31 January 2019 Accepted 2 February 2019 Available online 13 February 2019

Keywords: Natural capital Ecological capital Ecosystem services Ecological assets Virtual ecological capital

ABSTRACT

It is essential to assess the value of nature, as it provides various benefits for human economic development and well-being. It has been 20 years since two prominent publications came out in valuing nature. New concepts and methods have emerged since then. This study aimed to (1) investigate the relationship between the new proposed concept (ecological capital) and the existing two concepts: natural capital and ecosystem services and (2) examine the research trends of ecological capital accounting publications from 1997 to 2017. Bibliometric analysis was used to reveal the research trends. The results showed that the total number of publications has rapidly increased since 1997 and this growth trend will be maintained in the future. The most productive journal, country, institute, and author were Ecological Economics, USA, Chinese Academy of Sciences, and Dr. Verburg, respectively. Ecosystem services and land were the most frequent types of ecological capital. Non-economic evaluation approaches became less of a favorite over time and economic valuation methods were mostly applied in last five years. The integration of different methods has attracted increasing academic attention. The progress, advantages, and limitations of different methods were summarized in this study, including SEEA, ecological footprint, exergy, emergy, LCA, and economic valuation approaches, as well as newly developed modelling approaches. The last part of this study presented three challenges in this academic field – the need to (1) establish a standard framework, (2) consider the transfer/transport of ecological capital, and (3) improve capabilities for decision-making.

© 2019 Elsevier Ltd. All rights reserved.

1. Introduction

Along with rapid economic development and urbanization, resource shortages (Adamowicz et al., 2016), biodiversity loss (Shepherd et al., 2016), ecological degradation (Ouyang et al., 2016) and other ecological problems have frequently occurred. Human society gradually recognized nature's contribution in supporting the economy and human well-being (Tallis and Lubchenco, 2014; Chan et al., 2016). The value of nature is reflected in two aspects: (1) it provides direct resources, such as food, water, and timber and (2) it delivers indirect services, including carbon sequestration, pollination, recreation, and cultural values (MEA, 2005; Costanza et al., 2017). To sustain human society's development, nature as an important capital or asset must be effectively managed (Barbier, 2014; Polasky et al., 2015). Accounting for its value is the basis for nature's effective management. In 1997, the article – The value of the world's ecosystem services and natural capital was published in Nature (Costanza et al., 1997), as well as an edited book (Daily, 1997). These two prominent publications stimulated an explosion of research related to valuing nature (Costanza et al., 2017). For instance, three global initiatives, the Millennium Ecosystem Assessment (MEA) project, the Economics of Ecosystems and Biodiversity (TEEB) project, and the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services (IPBES) were successively established to evaluate the values of nature and mainstream them into decision-making at all levels (Costanza et al., 2017).

In the twenty years of development, some new concepts have been proposed, such as ecological capital (Barbier, 2016), ecological

^{*} Corresponding author. Department of Environmental Science and Engineering, Fudan University, 2005 Songhu Road, Yangpu District, Shanghai, 200438, China. E-mail address: yutaowang@fudan.edu.cn (Y. Wang).

assets (Galli et al., 2015) and ecosystem assets (Hein et al., 2016). And the accounting techniques have gradually expanded, previously they were limited to the ecological footprint (EF) and some economic valuation approaches, such as avoided cost, travel cost, and hedonic pricing (Costanza et al., 2017). Currently there are many cases of using benefit transfer, emergy, life-cycle assessment (LCA) and dynamic modelling (e.g., InVEST) to assess the value of nature. Costanza et al. (2014) estimated the global changes of values of ecosystem services through benefit transfer. Wang et al. (2019) applied emergy to account for the value of several key ecosystem services in the Yellow River Delta region in China. Liu et al. (2018a, b) developed a conceptual framework to evaluate ecosystem services with LCA. Hoyer and Chang (2014) modeled water related ecosystem services of a river basin, such as water yield and water retention with InVEST software. Many papers have summarized the progress in the field of assessing nature's values. For the relationship of concepts, Smith et al. (2017) synthesized a typology to link natural capital and ecosystem services. Maseyk et al. (2017) introduced a framework to illustrate how natural capital delivers ecosystem services for decision makers. For the development of accounting methods, Pascual et al. (2010) introduced the strengths, limitations and real-world cases of various economic valuation approaches. Richardson et al. (2015) described the role of the benefit transfer technique in facilitating the economic valuation of ecosystem services. Christie et al. (2012) compared the applicability of both economic and non-economic methods in valuing ecosystem services. Emerging methods such as remote sensing (de Araujo Barbosa et al., 2015) and dynamic modelling (Shoyama et al., 2017) have also been reviewed.

However, few studies have explored the relationship between the newly proposed concept (ecological capital) and the existing concepts (natural capital and ecosystem services). Clarifying the differences of these concepts is essential in conducting value assessments of nature. Because the three concepts cover different types of natural resources, the selection of valuation methods needs to be determined according to the specific resource type. Additionally, little research has analyzed the research trends of evaluation approaches from a bibliometric perspective. Bibliometric analysis is a common tool used to investigate the research patterns of a given field (Fu and Ho, 2013; Mao et al., 2015). It has been used in many disciplines, such as green supply chain management (Fahimnia et al., 2015), circular economy (Geissdoerfer et al., 2017), and water footprint research (Zhang et al., 2017).

Therefore, the objectives of this study were to (1) investigate the relationship of the three concepts: natural capital, ecosystem services and ecological capital or assets, and (2) examine the research trends in ecological capital accounting from 1997 to 2017 with bibliometric methods. This paper is as follows: Section 2 investigates the relationship of the three concepts. Section 3 introduces the methodology of this study. Section 4 presents the results of bibliometric analysis. Section 5 summarizes the development of each method. Section 6 presents the identified research challenges and section 7 draws the conclusions.

2. Concepts

Natural capital and ecosystem services concepts are at the core of literature of valuing nature. Their respective definitions and evaluation methods have been well examined, as has been the relationship between them. In last ten years, scientists have proposed some new concepts, such as ecological capital (Barbier, 2016), ecological assets (Galli et al., 2015) and ecosystem assets (Hein et al., 2016). These terms are collectively named as ecological capital in this study. There are overlaps and differences between the two existing concepts and the newly proposed one. This section

will briefly introduce the definitions of the three concepts and clarify their relationship, which helps in the choice of appropriate assessment methods.

2.1. Natural capital

The term "natural capital" has a long history in the literature. Its origin and development can be found in (Missemer, 2018) and (DesRoches, 2015). The classical definition of natural capital is a stock able to produce a flow of valuable goods or services into the future, and two groups of natural capital can be differentiated: (1) nonrenewable natural capital, refers to nonrenewable or exhaust-ible resources, such as fossil fuels and metal ores, and (2) renewable natural capital, including renewable resources (land, water, air, forest, and others) and ecosystems that can provide flows of ecosystem services (Costanza and Daly, 1992). This stock-flow definition was widely accepted by subsequent researchers (Häyhä and Franzese, 2014; Mancini et al., 2017).

2.2. Ecosystem services

"Ecosystem services" is another core concept that is often mentioned in the literature of valuing nature. Gómez-Baggethun et al. (2010) and Braat and de Groot (2012) summarized its developmental history in economics and other disciplines respectively. The MEA report defined ecosystem services as the benefits humans directly and indirectly acquire from ecosystems, and classified various ecosystem services into four groups: (1) provisioning services, products or goods directly gained from nature, such as food, timber, and biochemical, (2) regulating services, such as air quality regulation, water purification, and erosion control, (3) cultural services are the nonmaterial goods and services attained from nature, including sense of place, recreation and spiritual experience, and (4) supporting or habitat services, which support the operation of all other groups of services (MEA, 2005). Most subsequent studies have adopted this definition and classification, with minor changes in category names and attribution of some individual services (Schröter et al., 2014; Pascual et al., 2017). Some studies proposed other classification system. Wallace (2007) developed a typology classification framework that links human value and ecological structures and processes. Haines-Young and Potschin (2012) and Landers and Nahlik (2013) introduced a paradigm that sorts ecosystem services into supporting and final services. Fisher et al. (2009) suggested a classification scheme under the decision-making context.

2.3. Ecological capital

In addition to the above two concepts, some new terms have been proposed by researchers in the past decade. Barbier (2013, 2016) introduced "ecological capital" and considered that it has three characteristics: (1) depreciation, (2) irreplaceability; and (3) the ability to abruptly collapse. Galli et al. (2015) and Mancini et al. (2017) used the term "ecological assets", defined as "the spatial areas with biotic and abiotic components functioning together". Wang et al. (2011) proposed "natural assets" as the combination of five components: water, land, air, living organisms, and mineral resources. Hein et al. (2016) proposed "ecosystem assets", which include ecosystems capacity and capability to supply ecosystem services, as well as the potential supply of ecosystem, and natural assets are collectively referred as ecological capital in this study.

2.4. The relationship of the three concepts: natural capital, ecological capital, and ecosystem services

According to their classic definitions, ecosystem services can be understood as the flows of services generated by natural capital (Costanza and Daly, 1992). This idea is widely accepted in current mainstream research (Maseyk et al., 2017; Smith et al., 2017). Some researchers tend to consider ecological capital as the stock part (physical forms) of natural capital (Galli et al., 2015; Mancini et al., 2017), while others view it as including the flow part (ecosystem services) of natural capital, which is equivalent to natural capital (Frazier et al., 2013; Hein et al., 2016). There are also some that consider that non-renewable resources should be excluded (Wang et al., 2011; Barbier, 2013). The disputes are mainly concentrated in two aspects: (1) whether it should contain the flow component and (2) whether it should contain non-renewable resources.

To solve this controversy, we suggest that ecological capital should contain the flow part and exclude non-renewable resources. The relationship of the three concepts is shown in Fig. 1. There are several reasons for proposing such a description. First, it follows the classic definition of "capital" in natural capital, which was defined as a stock that able to generate flows. Second, "ecological" is a derivative of "ecosystem". Non-renewable resources are typically reserved to the lithosphere, which are not included in ecosystem areas in terms of geographical distribution. Finally, this definition distinguishes ecological capital from natural capital, avoiding the use of two phrases expressing the same meaning and splitting relevant resources.

3. Methodology

This section introduces the methods used in this study, as well as the data source and cleaning processes. To predict the trend of research outputs, the Hurst exponent and Mann-Kendall test were used, and the resulting H and Z values reflect the relative tendency of time-series data (Wang et al., 2017). The total available range for the H value is 0–1. An H value that falls between 0.5 and 1 indicates a strong propensity to an increasing trend, meaning that a high value in the series will probably be followed by another high value. A value in the range 0–0.5 indicates the time series is prone to return to the average, meaning high and low values adjacently appear. When H is 0.5, the time series is a completely uncorrelated series. For the Mann-Kendall test, a Z < 0 value means that the time series tends to decrease, while Z > 0 indicates an increasing trend.

3.1. Bibliometric analysis

Bibliometric analysis, originally introduced by Pritchard, is a collection of mathematical and statistical methods used to analyze written publications (Ellegaard and Wallin, 2015; Ellegaard, 2018).



Fig. 1. The relationship of natural capital, ecological capital, and ecosystem services: A - from the stock-flow perspective, adapted from (Costanza et al., 2017) and B - from the renewable and non-renewable resource perspective, adapted from (Liu et al., 2018).

It has become a typical and effective tool for systematic review in a variety of research fields, such as low carbon technologies (Wei et al., 2018), circular economy (Geissdoerfer et al., 2017), green supply chain (Fahimnia et al., 2015), and water footprint (Zhang et al., 2017). Bibliometric analysis helps to reveal the characteristics, structures and patterns of a particular topic, as well as research hotspots and future trends (Ellegaard and Wallin, 2015). A typical bibliometric analysis firstly evaluates the performance of selected publications, including distribution of various journals, countries, institutes, and authors, then investigates research hotspots and predict future research directions with keywords analysis (Mao et al., 2015; Mo et al., 2018). Two common indicators, the impact factor (IF) and h-index were used to evaluate the performance of selected publications (Zhong et al., 2016). The IF of an academic journal is frequently used as a proxy for the relative importance of a journal within its field; journals with higher impact factors are often deemed to be more important than those with lower ones. IF have been calculated yearly starting from 1975 for journals listed in the Journal Citation Reports. The h-index is an indicator that attempts to measure both the productivity (quantity) and citation impact (quality) of the publications of a scientist, an organization or a journal. Specifically, a higher h-index usually means a higher research performance. To facilitate the analysis, Bibexcel (Persson et al., 2009) was used to extract certain information of selected publications, such as author, published year, journal, address, citations, and keywords.

3.2. Social network analysis

Social network analysis (SNA) is the process of investigating social structures through the use of networks and graph theory (McLinden, 2013; Zhong et al., 2016). It characterizes networked structures in terms of nodes and edges. The nodes can be authors, research institutes, universities, and countries, and the edges or links represents their co-occurrences. SNA is an effective way to reflect the relationships among different actors (McLinden, 2013; Mao et al., 2018). Bibexcel was employed to extract relevant data and construct co-occurrence matrices, including author, keywords, address, journal title, citation and other information. The matrices were then treated by Pajek, VOSviewer and Gephi software to produce the co-occurrence network and clusters (Mao et al., 2015; Wang et al., 2017). The resulting graphs visualize the academic collaborations between authors, institutes, and countries.

3.3. Data collection

The following keywords were selected to search publications in the database of the Web of Science Core collection: ("ecosystem services" OR "natural capital" OR "ecological capital" OR "ecological assets") AND ("valuation" OR "assessment" OR "accounting"). The database covers the Science Citation Index Expanded (SCI-EXPANDED), Social Sciences Citation Index (SSCI), Conference Proceedings Citation Index- Science (CPCI–S), and Conference Proceedings Citation Index- Social Science & Humanities (CPCI-SSH) four key databases. The time interval was set from 1997 to 2017 because this field started to gain mainstream research interests in 1997. A total of 6163 records were found, consisting of 6114 (99.2%) English publications and 49 (0.8%) other language publications. Since English is the dominant academic language, the 6114 English documents were collected for further analysis.

4. Results

Among the 6114 English publications, articles (80.91%) were the dominant document type, followed by proceeding papers (8.41%),

reviews (7.92%), and other types of publications (1.76%). Fig. 2 shows the number of total publications (TP) per year between 1997 and 2017. The TP has rapidly grown, from 6 publications to approximately 1200. The H value was 0.9496, and the Z value was 6.13 > 0, indicating that the publications of this field will keep growing in the future.

4.1. Journal performance

Table 1 shows the top 20 most productive journals, which accounted for 37.88% of the total publications. *Ecological Economics* was the most productive journal (347) and ranked 1st in the h-index (67), indicating its dominance both in quantity and quality in this field. *Ecosystem Services* as a newly established journal (in 2012) achieved a comprehensive performance, ranked 2nd in publications (341) and 4th in the h-index (30), as well as a high impactor factor (4.395). *Sustainability* had a high quantity of publications (103), but with a relatively low impact factor (2.075) and h-index (12) values. Journals such as *PNAS*, *Journal of Applied Ecology*, and *Landscape and Urban Planning* had a low number of total publication (53, 58, and 81, respectively) yet got high impactor factors (9.504, 5.742, and 4.994, respectively) and h-index values (36, 26, and 29, respectively). The results reflect that diverse indicators are needed to represent a journal's performance.

4.2. Country performance and collaboration

Among the 6114 records, 6092 documents have affiliation information. The 6092 publications were collected for country and institute performance analysis. Results showed that 151 countries/ regions contributed to the ecological capital accounting literature. Table 2 lists the top 20 most productive countries. Their publications account for 83.14% of all documents. Among them, 13 countries are from Europe, with a share of 40.91%. The USA (1,919) is the most productive country, as well as in categories of first-author country and publications with and without international collaboration. It also ranked 1st in the h-index (119), indicating USA's dominance in this research area. The following most productive countries are the UK (1,073), Germany (689), and China (662), with a huge gap compared to the USA. The UK and Germany took 2nd and 3rd place in the h-index. In contrast, China merely ranked 13th in h-index, indicating its publications received less academic attention. The reason that China obtained a low h-index could be lack of international collaboration. Table 2 shows that the cooperation ratio of Chinese publications was C = 35.5%, much lower compared to other countries. International collaboration plays an important role in enhancing academic influence. Countries such as



Fig. 2. The performance of ecological capital accounting publications from 1997 to 2017.

Australia and Netherlands with higher levels of international collaboration achieved higher h-index values.

Fig. 3 presents the academic collaboration relationship among the 25 most productive countries. The Louvain method was applied in Pajek software to make a cluster analysis. Two clusters were identified, and the color of the nodes indicates which cluster it belongs to. The size of the nodes represents the number of a country's publications with international collaboration. The width of the edges refers to the frequency of collaboration between two countries. The USA occupies the core of the network and has the most cooperative publications. The connection between USA and UK is the strongest, with 233 publications. Canada, Australia and China are also closely linked to the USA, with 150, 129, and 120 coauthored publications, respectively. The UK is the most active country of the blue cluster, it had 144, 136, and 111 cooperative publications with Germany, Netherlands, and France, respectively. The blue cluster mainly consisted of European countries, indicating distance could be an important factor affecting international cooperation.

4.3. Institute performance and collaboration

Table 3 shows the top 15 most productive institutes and their performance. Seven institutes are in the USA, reflecting USA's leading position in this academic field. The Swedish University of Agricultural Sciences (SLU) is the only research organization not from the top 5 productive countries. The Chinese Academy of Sciences (CAS) is the most productive academic institute, with a total of 250 publications, as well as in first author publications and publications with and without cooperation. However, its h-index (31) ranked 10th. Another Chinese research institute, Beijing Normal University, had a similar performance to CAS. It ranked 5th in number of publications yet 47th in the h-index. The results indicated that their publications attracted fewer academic interests. The following most productive academic institutes are Wageningen University & Research, Helmholtz Centre for Environmental Research (UFZ), and Stanford University. All of them achieved high h-index performances, with a value of 53, 38, and 47, respectively.

Fig. 4 shows the academic collaboration network among the 20 most productive institutes. These institutes were divided into 3 clusters by the Louvain method in Pajek. The size of the nodes refers to the number of publications with collaboration, and the width of the lines refers to the frequency of collaboration between two institutes. Research organizations from the same continent are clustered together, red for North America, blue for Europe, and green for China. The CAS is the most active institute with 201 cooperative documents. Beijing Normal University is CAS's closest partner. They have published 20 documents together. Wageningen University & Research is the second most active research organization, it has a close connection with UFZ (16), Vrije University Amsterdam (13), and SLU (13). Stanford University – University of Minnesota and University of Cambridge – University of East Anglia are the most cooperative pairs, with 25 co-authored publications.

4.4. Author performance and collaboration

More than 20,000 authors have published research in this field. Fig. 5 presents the performance of authors who have published more than 20 research papers, a total of 18 authors. These authors are from 12 different countries, indicating this topic has attracted extensive interests from various nations. The grey dashed line helps to understand these authors' performance. The closer the author to the line, the better performance the author has achieved in both quality and quantity. Dr. Lavorel, Dr. Pascual, Dr. Chan, and Dr. Haase

Table 1

The top 20 most productive journals and their performance during 1997–2017.

Journal title	ТР	% (R)	IF-2017	h-index (R)
Ecological Economics	347	5.68 (1)	3.895	67 (1)
Ecosystem Services	341	5.58 (2)	4.395	30 (4)
Ecological Indicators	231	3.78 (3)	3.983	40 (2)
Journal of Environmental Management	126	2.06 (4)	4.005	27 (7)
Ecology and Society	123	2.01 (5)	3.256	29 (5)
Land Use Policy	116	1.90 (6)	3.194	24 (13)
PLOS One	115	1.88 (7)	2.766	25 (10)
Science of the Total Environment	113	1.85 (8)	4.61	23 (14)
Sustainability	103	1.68 (9)	2.075	12 (43)
Environmental Science & Policy	88	1.44 (10)	3.826	25 (10)
Landscape and Urban Planning	81	1.32 (11)	4.994	29 (5)
Ecological Modelling	72	1.18 (12)	2.507	21 (18)
Agriculture Ecosystems & Environment	66	1.08 (13)	3.541	25 (10)
Environmental Management	65	1.06 (14)	2.177	18 (24)
Biological Conservation	58	0.95 (15)	4.66	27 (7)
Journal of Applied Ecology	58	0.95 (15)	5.742	26 (9)
Landscape Ecology	56	0.92 (17)	3.833	21 (18)
PNAS ^a	53	0.87 (18)	9.504	36 (3)
Ocean & Coastal Management	53	0.87 (18)	2.276	15 (33)
Regional Environmental Change	51	0.83 (20)	2.872	14 (35)

TP: total publication; % (R): the ratio and rank of the publications; IF-2017: the impactor factor of the journal in 2017; h-index (R): h-index and its rank; a: PNAS – Proceedings of the National Academy of Sciences of the United States of America.

Table 2

The top 20 most productive countries during 1997-2017.

Country	TP	% (R)	FP% (R)	SP% (R)	CP%(R)	C%	h-index (R)
	1010	21 20 (1)	22 10 (1)	16.9 (1)	14.62 (1)	46.6	110 (1)
	1919	1755(1)	22.19(1)	10.0(1)	14.02 (1)	40.0	119(1)
UK Communit	1075	17.55 (2)	9.42 (5)	0.2(3)	11.55 (2)	04.0	89 (Z) 72 (2)
Germany	689	11.27 (3)	5.82 (4)	3.6 (4)	7.65 (3)	67.9	72(3)
China	662	10.83 (4)	9.57 (2)	7.0(2)	3.84 (10)	35.5	45 (13)
Australia	511	8.36 (5)	5.36 (5)	3.1 (5)	5.27 (5)	63.0	63 (5)
Netherlands	446	7.29 (6)	2.75 (10)	1.6 (10)	5.66 (4)	77.6	67 (4)
Italy	440	7.20 (7)	4.22 (6)	2.8 (6)	4.37 (8)	60.7	47 (10)
France	422	6.90 (8)	3.83 (8)	1.8 (7)	5.14 (6)	74.4	53 (8)
Spain	405	6.62 (9)	3.97 (7)	1.7 (9)	4.94 (7)	74.6	58 (6)
Canada	355	5.81 (10)	3.04 (9)	1.8 (7)	4.04 (9)	69.6	54(7)
Sweden	295	4.82 (11)	2.00 (11)	1.3 (11)	3.53 (11)	73.2	53 (8)
Switzerland	263	4.30 (12)	1.83 (12)	0.8 (13)	3.48 (12)	81.0	46 (11)
South Africa	188	3.07 (13)	1.77 (13)	1.2 (12)	1.88 (15)	61.2	46 (11)
Denmark	164	2.68 (14)	1.42 (14)	0.4 (23)	2.31 (13)	86.0	35 (14)
Belgium	150	2.45 (15)	1.34 (16)	0.7 (18)	1.80 (16)	73.3	32 (16)
Finland	146	2.39 (16)	1.23 (17)	0.8 (15)	1.62 (18)	67.8	34 (15)
Brazil	146	2.39 (16)	1.42 (14)	0.8 (14)	1.59 (19)	66.4	31 (18)
Austria	144	2.36 (18)	0.93 (21)	0.4 (23)	1.98 (14)	84.0	32 (16)
Portugal	125	2.04 (19)	0.83 (22)	0.6 (20)	1.41 (20)	68.8	25 (24)
New Zealand	113	1.85 (20)	0.98 (19)	0.7 (17)	1.11 (23)	60.2	30 (19)

TP: total publication; %(R): the ratio and rank of the publications; FP% (R): the ratio and rank of first author publications; SP% (R): the ratio and rank of publications within international collaboration; C%: the ratio of international collaborative publications of a country; hindex (R): h-index and its rank.

have a better performance. If the author is under the dashed line, it means his/her publications received less citations.

Fig. 6 shows the collaboration network of authors who have published more than 14 articles. A total of 45 authors were selected. They were classified into 6 clusters through the Louvain method, and the color of the nodes refers to which cluster it belongs to. The size of the circles refers to the cooperative number of publications. The width of the links represents the frequency of cooperation between two researchers. The purple cluster represents a group of authors focusing on land related ecosystem services and land use modelling. Researchers belonging to the orange cluster are interested in biodiversity, conservation and ecosystem services, and they show a stronger connection compared to other clusters. Scholars with a blue color have a background of ecological economics, such as Dr. Costanza and Dr. de Groot, and they focus on economic valuation approaches in valuing nature. The light green scientists are interested in using Integrated Valuation of Ecosystem Services and Tradeoffs (InVEST) models to evaluate ecosystem services. For example, Dr. Daily is the major contributor of InVEST models. The dark green cluster represents two Chinese scholars, Dr. Fu and Dr. Lu. Land resource and land related ecosystem services are their main research interests. The pink cluster shows scientists developing Artificial Intelligence for Ecosystem Services (ARIES) and Social Values for Ecosystem Services (SoLVES) models.

4.5. Keywords analysis

Keywords reflect the research focus of a paper and help to identify the research trends of an area. All the keywords from the 6114 documents were extracted and pre-treated. For example, "economic valuation" represents "economic analysis", "economic assessment", and "monetary valuation". There are 65 keywords





Fig. 4. The academic collaboration network of the 20 most productive institutes during 1997–2017.

Fig. 3. The collaboration network of the 25 most productive countries during 1997–2017.

Table 3

The top 15 most productive institutes during 1997-2017.

Institute	TP (R)	FP % (R)	SP% (R)	CP% (R)	C%	h-index (R)
Chinese Acad Sci, China	250 (1)	2.58 (1)	0.80 (1)	3.29 (1)	80.40	31 (10)
Wageningen UR, Netherlands	210 (2)	1.21 (2)	0.43 (2)	3.01 (2)	87.62	53(1)
UFZ ^a , Germany	153 (3)	1.06 (3)	0.31 (5)	2.19 (3)	87.58	38 (4)
Stanford Univ, USA	126 (4)	0.79 (6)	0.16 (18)	1.90 (4)	92.06	47 (2)
US EPA, USA	108 (5)	0.95 (4)	0.39 (3)	1.37 (7)	77.78	25 (21)
Beijing Normal Univ, China	97 (6)	0.90 (5)	0.38 (4)	1.21 (16)	76.29	19 (47)
Univ Cambridge, USA	95 (7)	0.44 (22)	0.15 (25)	1.41 (5)	90.53	38 (4)
Stockholm Univ, USA	94 (8)	0.54 (12)	0.15 (25)	1.39 (6)	90.43	36 (6)
Univ E Anglia, UK	92 (9)	0.51 (15)	0.16 (18)	1.34 (9)	89.13	41 (3)
Vrije Univ Amsterdam, Netherlands	92 (9)	0.69 (8)	0.23 (8)	1.28 (12)	84.78	36 (6)
US Forest Serv, USA	92 (9)	0.57 (10)	0.15 (25)	1.36 (8)	90.22	27 (18)
US Geol Survey, USA	91 (12)	0.70(7)	0.16 (18)	1.32 (10)	89.01	29 (13)
Univ Minnesota, USA	89 (12)	0.46 (19)	0.16 (18)	1.29 (11)	88.76	36 (6)
Swedish Univ Agr Sci, Sweden	87 (14)	0.51 (15)	0.23 (8)	1.19 (17)	83.91	23 (28)
Univ British Columbia, Canada	86 (15)	0.54 (12)	0.16 (18)	1.24 (15)	88.37	30 (12)
US EPA, USA Beijing Normal Univ, China Univ Cambridge, USA Stockholm Univ, USA Univ E Anglia, UK Vrije Univ Amsterdam, Netherlands US Forest Serv, USA US Geol Survey, USA Univ Minnesota, USA Swedish Univ Agr Sci, Sweden Univ British Columbia, Canada	108 (5) 97 (6) 95 (7) 94 (8) 92 (9) 92 (9) 92 (9) 91 (12) 89 (12) 87 (14) 86 (15)	$\begin{array}{c} 0.95\ (4)\\ 0.90\ (5)\\ 0.44\ (22)\\ 0.54\ (12)\\ 0.51\ (15)\\ 0.69\ (8)\\ 0.57\ (10)\\ 0.70\ (7)\\ 0.46\ (19)\\ 0.51\ (15)\\ 0.54\ (12) \end{array}$	$\begin{array}{c} 0.39\ (3)\\ 0.38\ (4)\\ 0.15\ (25)\\ 0.15\ (25)\\ 0.16\ (18)\\ 0.23\ (8)\\ 0.15\ (25)\\ 0.16\ (18)\\ 0.23\ (8)\\ 0.16\ (18)\\ 0.23\ (8)\\ 0.16\ (18)\\ 0.23\ (8)\\ 0.16\ (18)\\ 0.23\ (8)\\ 0.16\ (18)\ (18)\ $	$\begin{array}{c} 1.37\ (7)\\ 1.21\ (16)\\ 1.41\ (5)\\ 1.39\ (6)\\ 1.34\ (9)\\ 1.28\ (12)\\ 1.36\ (8)\\ 1.32\ (10)\\ 1.29\ (11)\\ 1.19\ (17)\\ 1.24\ (15) \end{array}$	77.78 76.29 90.53 90.43 89.13 84.78 90.22 89.01 88.76 83.91 88.37	25 (21) 19 (47) 38 (4) 36 (6) 41 (3) 36 (6) 27 (18) 29 (13) 36 (6) 23 (28) 30 (12)

TP: total publication and its rank; FP% (R): the ratio and rank of first author publications; SP% (R): the ratio and rank of publications without collaboration; CP% (R): the ratio and rank of publications with collaboration; C%: the ratio of collaboration publications of an institute; h-index (R): h-index and its rank; ^a UFZ: Helmholtz Centre for Environmental Research.

that appeared more than 100 times. Three categories of keywords, research location, type of resources, and methods, were then collected to analyze research trends. As shown in Fig. 7, there are about 1400 keywords referring to research locations, among which 28.9% are European countries. The rest are the USA (22.8%), China (14.3%), Australia (6.1%), UK (5.0%), and other countries.

There were 7182 keywords focusing on types of resources, a quarter of which were ecosystem services. The following were land (22.3%), biomass or biodiversity (16.3%), forest (15.2%), water (11.5%), and others. Fig. 8 represents the time variation (five-year interval) of research focus in types of resources. The frequency of each resource had an upward trend. In terms of ratio, most resources fluctuated with time. Only land showed an increasing trend, and the share of ecosystem services and water decreased over time. Method-related keywords analysis is presented in next section.



Fig. 5. The performance of the top 18 productive authors during 1997-2017.



Fig. 6. The collaboration network of authors (TP > 14) during 1997–2017.



Fig. 7. The distribution of research focus during 1997–2017.

5. Measuring ecological capital: state of the art and trends

Ecological capital contains two components: renewable natural capital stock and ecosystem services. Each part has a variety of frameworks and methods, and the value of ecological capital can be expressed in physical and monetary terms. This section presents the trend of commonly used methods, and summaries the developments of each, including integrated Environmental and



Fig. 9. The application of various methods during 1997–2017.

Economic Accounting (SEEA), EF, exergy, emergy, and LCA for natural capital accounting, and biophysical and monetary value evaluation approaches for ecosystem services.

5.1. Trends

There were approximately 2131 method-related keywords, and their distribution is showed in Fig. 9. Economic and non-economic approaches each accounted for about 50%. The stated preference method was the most popular approach of economic valuation approaches, with 599 occurrences. The contingent valuation method (CVM) and choice experiment were the two most applied stated preference methods, with 292 and 117 occurrences, respectively. In terms of non-economic or biophysical evaluation methods, model-related methods occurred 368 times, including InVEST, SoLVES, and ARIES. LCA took the second position, accounting for 15.9%. The following methods were EF (12.3%), emergy (11.5%), spatial analysis (9.8%), GIS (9.2%), exergy (4.3%), and others.

Fig. 10 reflects the frequency of occurrence and the corresponding ratio of method-related keywords with time. From the ratio perspective, EF was the most popular approach, yet its percentage gradually decreased and currently is almost the least used one. Economic valuation approaches remained a dominant method and their proportion has gradually increased. Presently, it is the favorite approach. The ratio of models and LCA both increased over time and are the 2nd and 3rd used methods in last five years. The share of emergy and exergy had a decreasing trend. Spatial analysis first appeared in 2008 and its ratio has been increasing since then. Another space-related method, GIS has a similar trend. There are growing numbers of studies applying GIS or spatial analysis in measuring ecological capital. de Araujo Barbosa et al. (2015) and Shoyama et al. (2017) summarized the remote sense and modelling approaches and practical cases.



Fig. 8. The time variation of focused types of resources during 1997-2017.



Fig. 10. The time variation of method-related keywords during 1997-2017.

Ecological capital encompasses diverse resources and services, and each of them has a relatively appropriate accounting approach. To evaluate the true value of ecological capital, different methods are needed, and the combined use of methods has become a trend. Reza et al. (2014) explored the possibility of integrating emergy into LCA and calculated the impacts of paved roads on ecosystems. Wang et al. (2019) assessed several key ecosystem services in China through the incorporation of modelling, GIS, and emergy, Fig. 11 represents the combination of various approaches based on their co-occurrence. The size of the nodes refers to the total number of co-occurrence of one method, and the width of the edges indicates their frequency of co-occurrence. These techniques are divided into two clusters by Gephi. The blue cluster represents methods that are usually applied to evaluate natural capital, while the green cluster stands for ecosystem valuation approaches. Models are the most favored approach for integration. InVEST models typically involve GIS inputs and spatial analysis (Goldstein et al., 2012; Arkema et al., 2015). ARIES and SoLVES models often utilize GIS to show the spatial distribution of economic valuation results (Sherrouse et al.,



Fig. 11. The co-occurrence of different methods during 1997-2017.

2014; Villa et al., 2014). Emergy, exergy and EF use coefficients to transform into each other, and their values can be treated as inputs into LCA (Zhang et al., 2010b; Raugei et al., 2014). Consequently, these approaches are often combined.

5.2. Natural capital accounting

Natural capital consists of two parts, the stock and flow components. This subsection introduces the common methods used to evaluate the stock part. Ecosystem services valuation approaches are presented in next subsection. The focus of this paper is on ecological capital, so accounting approaches for renewable resources are mainly discussed.

5.2.1. Integrated Environmental and Economic Accounting (SEEA)

The SEEA is a framework for accounting for national natural resources. Natural resources play an essential role in a country's economic development, and thus their stock and flow changes should be documented (Obst et al., 2016). The first handbook of SEEA was released in 1993. It was continuously revised since then, and the latest manual was published in 2012 (Bartelmus, 2014; UN, 2014a). Many nations such as Australia (Obst and Vardon, 2014), Netherlands (Remme et al., 2015) have adopted SEEA framework. The SEEA framework has also been applied to account for specific types of sources, such as water (Borrego-Marín et al., 2016; Pedro-Monzonís et al., 2016) and soil (Robinson et al., 2014). As a satellite account to the System of National Accounts (SNA), SEEA connects the economic system with the environment. However, it is not clear how specific types of natural resources circulate in the socioeconomic system. Some researchers constructed input-output database to analyze resource metabolism in society (Wood et al., 2015; Stadler et al., 2018).

One problem with the old SEEA framework is that it separately accounts for different resources. Each type of resources is expressed in its physical value, such as hectares for land resource (Bartelmus, 2014). The revised version incorporated economic valuation methods, so that all categories of resources can be converted into and summed up in monetary value, which helps to facilitate the inclusion of assessment results in decision-making (Galos et al., 2015; Vardon et al., 2016). Another limitation of SEEA is that the indirect value of nature such as climate regulation, water purification and pollination services (Obst et al., 2016) were neglected in the framework. These indirect benefits provide significant support

for social and economic development. The SEEA Experimental Ecosystem Accounting (SEEA-EEA) was developed for the valuation of non-market values of nature in 2014 (UN, 2014b). Lai et al. (2018) adopted the SEEA-EEA framework and quantified water-related services in Finland. La Notte et al. (2017) conducted a case study of assessing the value of the nitrogen retention service for Europe.

5.2.2. Ecological footprint

The EF is a method proposed in 1990s to measure the land demand of human activities against the supply of the earth (Wackernagel and Rees, 1997; Wackernagel et al., 2018b). If the demand exceeds supply, the targeted region is subjected to an ecological deficit (Collins et al., 2018). After more than 20 years of development, the EF has been gradually improved and widely used in different geographic scales (Wackernagel et al., 2018a). The Global Footprint Network (GFN) established the national footprint accounting (NFA) framework and publishes national footprint results at intervals. According to their latest research, the footprint of humanity greatly surpassed the capacity of earth, with an additional 0.7 earth needed for the future (Lin et al., 2018). EF has become a favorite tool to assess the ecological supply and demand at different geographic scales. Galli et al. (2017) applied inputoutput models to assess the food sector's ecological footprint of 15 Mediterranean countries. Rugani et al. (2014) accounted for the change of ecological capital in Luxembourg from 1995 to 2009 through hybrid input-output models. Baabou et al. (2017) used a top-down EF approach and estimated the ecological footprint of 19 Mediterranean cities.

Some aspects of the EF have been often criticized. On one hand, the equivalence factor used in the conventional EF ignores the multiple uses of land (Galli et al., 2016). Venetoulis and Talberth (2008) proposed calculating the equivalent factor based on net primary productivity (NPP), which considers land occupation of other species. On the other hand, apart from land and carbon sequestration services, it does not take into account other resources and services. Some scholars have proposed the water footprint (Fang and Heijungs, 2015; Fang et al., 2016) and material footprint (Wiedmann et al., 2015) to evaluate other kinds of resources. Mancini et al. (2018) examined the potential of using the EF to value ecosystem services and compared the results against economic approaches. In addition, Niccolucci et al. (2009) presented a threedimensional ecological footprint (^{3D}EF), which can better express the deficit between the human demand and supply of earth. Cases that utilized the applied the ^{3D}EF model can be found in (Fang et al., 2018; Yang and Hu, 2018).

5.2.3. Exergy analysis

Exergy is generally understood as the available energy that can be used (Zhang et al., 2010b). In principle all kinds of resources can be transformed into exergy. One advantage of exergy is presenting the value of various resources with one metric (Zhang et al., 2010b; Zhong et al., 2016). Exergy analysis can help to improve the efficiency of a process. It is typically used to assess the inputs of nonrenewable resources, such as metal ores, fossil fuels and others. Valero et al. (2018) applied exergy to assess the degradation of mineral resources. Some researchers suggest considering ecological resources and processes in exergy analysis, such as ecological cumulative exergy consumption (Ukidwe and Bakshi, 2007) and ecoexergy (Jørgensen, 2010; Zhang et al., 2010a). These methods are quite similar to emergy analysis (Zhang et al., 2010b; Lu et al., 2015), which is illustrated in the next section.

5.2.4. Emergy analysis

Emergy analysis was proposed in the 1990s by the prominent ecologist H.T. Odum (Odum et al., 2000; Chen et al., 2016). Emergy

theory views the earth as a system and its primary energy source is solar energy. All kinds of resources and ecological processes are formed and driven by solar energy (Zhang et al., 2010b; Chen et al., 2016). Accordingly, all goods and services can be converted into solar energy equivalents based on the solar energy they consume. The conversion factor is defined as transformity and determined by the total solar energy absorbed by the earth. The annual total solar energy is the baseline, whose assessment is at the core of emergy theory (Brown et al., 2016). Scholars in the emergy community have continuously updated the baseline. The latest result is presented in (Brown et al., 2016; Campbell, 2016).

Compared to other methods, emergy has three advantages. First, various forms of materials can be expressed by one unit - solar emjoules (sej). Many studies have applied emergy to assess the total value of different types of resources. Vassallo et al. (2017) estimated the value of marine ecosystems. Wang et al. (2016) assessed the natural capital change in the Yellow River Delta region in China by the integration of GIS and emergy. Mellino et al. (2015) mapped the spatial distribution of the key natural capital in the Campania region, as well as man-made assets. Secondly, emergy value can be converted into monetary terms through the emergy-to-dollar coefficient (Campbell and Tilley, 2014a). Monetary value is generally easier to be understood for by decision makers. Finally, emergy is appropriate to evaluate some ecosystem services, including provisioning, supporting, and regulating services (Coscieme et al., 2014). Wang et al. (2019) accounted for the value of several key ecosystem services in the Yellow River Delta region in China, such as soil retention, water provision and carbon sequestration. Other applications of emergy for ecosystem services evaluation can be found in, forest ecosystem services (Campbell and Tilley, 2014b), water and carbon ecosystem services (Watanabe and Ortega, 2014), and flood control (Chang and Huang, 2015). Nevertheless, cultural services cannot be valued through emergy.

5.2.5. Life-cycle assessment

LCA is a common tool for decision-making, which accounts for the environmental impact of a product or service throughout its life-cycle (ISO, 2006). LCA has two limitations in assessing natural capital. On one hand, conventional LCA primarily considers the consumption of abiotic resources, such as fossil fuel, metal and non-metal minerals, and impacts on human (Dewulf et al., 2015; Schaubroeck and Rugani, 2017). Improvements have been made to assess the consumption of renewable resources and ecosystem services, as well as impacts on other species and the natural environment (Chaplin-Kramer et al., 2017). Müller-Wenk and Brandão (2010) developed a framework to calculate the land-use impact on climate regulation services through LCA. Koellner et al. (2013) established the LCA guidelines to assess the land-use impact on biodiversity and ecosystem services. Other methods to account for ecosystem services with LCA can be found in: water-related services (Saad et al., 2013) and soil erosion service (Quinteiro et al., 2014). Liu et al. (2018a, b) developed a conceptual framework to evaluate a variety of ecosystem services with LCA. Pizzirani et al. (2014) even explored the possibility of assessing cultural values with LCA. Nevertheless, most of these studies are at a preliminary stage, and more efforts are needed to incorporate renewable resources and ecosystem services into LCA.

5.3. Ecosystem services valuation

"Ecosystem services" is the flow component of ecological capital. Since ecological capital is renewable, it is difficult to directly account for its value. Researchers typically evaluate the value of its flow part as a proxy of ecological capital. This section reviews ecosystem services assessment approaches, including biophysical and monetary approaches.

5.3.1. Biophysical value evaluation approaches

Ecosystem services are produced and maintained by ecological structures and processes (MEA, 2005; Costanza et al., 2017). Some services can be measured through simulation models of corresponding ecological processes. InVEST is a software that integrates many ecosystem service models and has been widely applied (Goldstein et al., 2012; Guerry et al., 2015). Fu et al. (2014) evaluated the ecosystem services provided by hydropower plants in the Zagunao River Basin. Arkema et al. (2015) estimated the value of ecosystem services provided by marine ecosystems in Belize. Hoyer and Chang (2014) modeled water related ecosystem services of a river basin, such as water yield and water retention. Grafius et al. (2016) accounted for several land related ecosystem services in urban areas with carbon storage, soil retention and pollination models. Shoyama et al. (2017) summarized cases that used modelling approaches in the Asia region.

There are some limitations with modelling approaches. First, the relationship between some services and corresponding processes remains unclear (Costanza et al., 2017). These models need to be optimized. Efforts have been made in model optimization, such as the soil retention model (Hamel et al., 2015). A second shortcoming is that it requires large ecological data inputs, such as water, soil, air and geological information. These inputs require considerable field research, which is typically expensive and even difficult for some regions. Therefore, some studies use default values as substitutions. Third, cultural values cannot be measured through InVEST. Models such as SoLVES (Sherrouse et al., 2014) and ARIES (Villa et al., 2014) were developed to assess some cultural services. Nevertheless, the raw data inputs of the two models are obtained through economic methods.

5.3.2. Economic valuation approaches

Economic or monetary valuation approaches refer to a series of methods that values ecosystem services through economic means (Häyhä and Franzese, 2014; Costanza et al., 2017). It is the primary technique in ecosystem services evaluation. According to market settings, monetary valuation methods can be classified into three groups: (1) direct market valuation methods, (2) revealed preference approaches, and (3) stated preference approaches (TEEB, 2010; Costanza et al., 2017). Each category can be further divided, and precise classification can be found in (MEA, 2005; TEEB, 2010). There are so many types of ecosystem services, and each of them has relatively appropriate approaches (Díaz et al., 2015; Pascual et al., 2017). In general, provisioning services are better to use direct market valuation methods, while cultural services are commonly assessed with revealed or stated reference approaches (Häyhä and Franzese, 2014; Kenter et al., 2015).

Monetary methods have two strengths. First, the result is presented in monetary term, which can be better understood by decision makers (Costanza et al., 2017). However, some scholars hold a different view and argue that economic value is not objective and cannot reflect the true value of nature, because these methods heavily rely on human preferences (Schröter et al., 2014). The dependence on human preferences can in fact be viewed as an advantage, because this is how the cultural services are valued (Kenter et al., 2015). For example, Van Berkel and Verburg (2014) estimated the cultural value of an agricultural landscape through the willingness to pay (WTP) exercise. Castro et al. (2014) applied CVM to assess the value of several landscape interests in Spain and compared the results with biophysical valuation in order to investigate the disparity between the supply and demand of ecosystem services. Questionnaires are an expensive approach in terms of cost and time. Some researchers suggested adapting empirical valuation results from one place or time to another. This technique is defined as benefit transfer and has been broadly adopted. For example, the global value of ecosystem services was estimated through benefit transfer (Costanza et al., 2014). Frélichová et al. (2014) constructed a database that includes 190 values of ecosystem services and used the benefit transfer technique to estimate the ecosystem services value in Czech Republic. Chaikumbung et al. (2016) synthesized more than 1400 wetland ecosystem services valuation cases and calculated the benefit transfer function for wetland. Rchiardson et al. (2015) and Costanza et al. (2010) summarized each method's applicability in using benefit transfer.

6. Challenges

Ecological capital provides vital support for human development. Accounting for changes of its value helps achieve effective management of ecological capital. After 20 years' development, many achievements have been obtained, yet there are still some challenges that need to be overcome for further progress (Guerry et al., 2015; Costanza et al., 2017). This section presents the three primary challenges in ecological capital accounting.

The first barrier is to establish a standard framework for ecological capital accounting (Díaz et al., 2015; Pascual et al., 2017). The authors of this paper define ecological capital as the combination of renewable natural capital stock and ecosystem services. Both fields have various methods, and each approach has its own advantages and limitations in assessing specific types of ecological capital (Kenter et al., 2015). For example, different kinds of resources or services can be expressed in one metric with EF, emergy and exergy. Non-economic approaches including SEEA, EF, emergy, exergy, LCA and InVEST, are not appropriate to assess the cultural and social values of nature (Zhang et al., 2010a,b; Christie et al., 2012). To facilitate decision-making, monetary valuation approaches are more appropriate as their results are presented in monetary terms (Costanza et al., 2017). However, the objective of economic approaches is controversial as most of them rely on human preferences (Schröter et al., 2014). In practice, which method to choose depends on the targeted types of ecological capital. A standard framework can help to decide the most appropriate approach, as well as facilitate comparison of different cases (Costanza et al., 2017; Pascual et al., 2017).

Another challenge is the need to consider the transfer or transport of ecological capital in an open and complex system. Currently most studies assess ecological capital without considerations of its movement. Many types of renewable resources, such as food and water, are transported to different locations through trade. These kinds of ecological capital usually associated with industrial products, can be referred as virtual ecological capital. When assessing a territory's ecological capital, it is insufficient to only consider its native ecological capital. Import and export flows should be considered as well. Input-output analysis is a potential tool to investigate this issue (Wood et al., 2015; Stadler et al., 2018). Furthermore, most ecosystem services have spatial limitations in delivering benefits. It is important to explore their serviceshed.

The last challenge is how to better incorporate ecological capital assessment into the decision-making process (Schaefer et al., 2015; Schultz et al., 2015). From the supply side, ecological capital is produced and maintained by ecosystems, and an increase of one desired service can result in a decrease of other desired ones. The trade-offs among various resources and ecosystem services should be scrutinized (Howe et al., 2014; Lu et al., 2014). From the demand side, people value nature in different aspects. For example, some people cherish the conservation value of natural land, while local

farmers prefer to convert them into agricultural land for production. Opinions of various stakeholders shall be included in relevant policy-making (Mouchet et al., 2014). In addition, the relationship between supply and demand of ecological capital should be further investigated (Bagstad et al., 2014).

7. Conclusion

In 1997, two prominent publications stimulated an explosion of research interests in valuing nature. In the 20 years of development, some new concepts such as ecological capital emerged. The authors of this study suggest defining ecological capital as the integration of renewable natural resources (renewable natural capital stock) and ecosystem services. This definition clarifies its conceptual relationship with the two existing concepts: natural capital and ecosystem services.

Various methods have been applied in valuing nature. This study used bibliometric methods to analyze the research trends. The results showed that the total publications have rapidly grown, from 6 publications in 1997 to 1151 in 2017. The most productive journal, country, institute, and author were Ecological Economics, USA, CAS, and Dr. Verburg respectively. Chinese research organizations and scholars had a relatively poor performance in h-index. Lack of international cooperation is a possible reason. Keywords analysis showed that the most concerned types of ecological capital were ecosystem services and land resources. In terms of assessment methods, the share of non-economic approaches has decreased over time. EF was the favorite method in the early stages, yet it has become less and less popular. Economic valuation approaches have remained as a dominant approach and have become the most applied ones in last five years. Some new techniques have emerged, such as dynamic modelling: InVEST, ARIES, and SoLVES. The integration of GIS technology helps to visualize the spatial distribution of ecological capital. The combined use of several methods has gained increasing academic attention, which can overcome the limitations of using single approach.

Despite so many achievements, there are three main challenges that need to be overcome for further development. The first barrier is the need to establish a standard framework. Each approach has its own advantages and limitations in assessing specific types of ecological capital. A standard framework can help to decide the most appropriate approach. A second challenge is the need to consider the transfer of ecological capital between different regions. Multi-regional input-output tables (MRIO) is a potential method for investigating the transportation of ecological capital. Finally, to better support decision-making, the following three issues need be further examined: the trade-offs among various kinds of ecological capital, preferences of different people, and the disparity between the supply and demand of ecological capital.

Acknowledgements

This study was founded by National Natural Science Foundation of China (No.71774032) and Newton Advanced Fellowship from the British Academy (NAFR2180103). Thanks to Dr. Edward C. Mignot, Shandong University, for linguistic advice.

References

- Adamowicz, W.L., Adamowicz, W.L., Olewiler, N., 2016. Helping markets get prices right: natural capital, ecosystem services, and sustainability. Can. Public Policy 42, S32–S38.
- Arkema, K.K., Verutes, G.M., Wood, S.A., Clarke-Samuels, C., Rosado, S., Canto, M., Rosenthal, A., Ruckelshaus, M., Guannel, G., Toft, J., 2015. Embedding ecosystem services in coastal planning leads to better outcomes for people and nature. Proc. Natl. Acad. Sci. U.S.A. 112, 7390–7395.

- Baabou, W., Grunewald, N., Ouellet-Plamondon, C., Gressot, M., Galli, A., 2017. The Ecological Footprint of Mediterranean cities: awareness creation and policy implications. Environ. Sci. Policy 69, 94–104.
- Bagstad, K.J., Villa, F., Batker, D., Harrison-Cox, J., Voigt, B., Johnson, G.W., 2014. From theoretical to actual ecosystem services: mapping beneficiaries and spatial flows in ecosystem service assessments. Ecol. Soc. 19, 64.
- Barbier, E.B., 2013. Wealth accounting, ecological capital and ecosystem services. Environ. Dev. Econ. 18, 133–161.
- Barbier, E.B., 2014. Economics: account for depreciation of natural capital. Nature News 515, 32.
- Barbier, E.B., 2016. The protective value of estuarine and coastal ecosystem services in a wealth accounting framework. Environ. Resour. Econ. 64, 37–58.
- Bartelmus, P., 2014. Environmental–economic accounting: progress and digression in the SEEA revisions. Rev. Income Wealth 60, 887–904.
- Borrego-Marín, M.M., Gutiérrez-Martín, C., Berbel, J., 2016. Estimation of cost recovery ratio for water services based on the System of Environmental-Economic Accounting for Water. Water Resour. Manag. 30, 767–783.
- Braat, L.C., de Groot, R., 2012. The ecosystem services agenda: bridging the worlds of natural science and economics, conservation and development, and public and private policy. Ecosyst. Serv. 1, 4–15.
- Brown, M.T., Campbell, D.E., De Vilbiss, C., Ulgiati, S., 2016. The geobiosphere emergy baseline: a synthesis. Ecol. Model. 339, 92–95.
- Campbell, D.E., 2016. Emergy baseline for the Earth: a historical review of the science and a new calculation. Ecol. Model. 339, 96–125.
- Campbell, E.T., Tilley, D.R., 2014a. The eco-price: how environmental emergy equates to currency. Ecosyst. Serv. 7, 128–140.
- Campbell, E.T., Tilley, D.R., 2014b. Valuing ecosystem services from Maryland forests using environmental accounting. Ecosyst. Serv. 7, 141–151.
- Castro, A.J., Verburg, P.H., Martín-López, B., Garcia-Llorente, M., Cabello, J., Vaughn, C.C., López, E., 2014. Ecosystem service trade-offs from supply to social demand: a landscape-scale spatial analysis. Landsc. Urban Plann. 132, 102–110.
- Chaikumbung, M., Doucouliagos, H., Scarborough, H., 2016. The economic value of wetlands in developing countries: a meta-regression analysis. Ecol. Econ. 124, 164–174.
- Chan, K.M., Balvanera, P., Benessaiah, K., Chapman, M., Díaz, S., Gómez-Baggethun, E., Gould, R., Hannahs, N., Jax, K., Klain, S., 2016. Opinion: why protect nature? Rethinking values and the environment. Proc. Natl. Acad. Sci. U.S.A. 113, 1462–1465.
- Chang, L., Huang, S., 2015. Assessing urban flooding vulnerability with an emergy approach. Landsc. Urban Plann. 143, 11–24.
- Chaplin-Kramer, R., Sim, S., Hamel, P., Bryant, B., Noe, R., Mueller, C., Rigarlsford, G., Kulak, M., Kowal, V., Sharp, R., 2017. Life cycle assessment needs predictive spatial modelling for biodiversity and ecosystem services. Nat. Commun. 8, 15065.
- Chen, D., Liu, Z., Luo, Z., Webber, M., Chen, J., 2016. Bibliometric and visualized analysis of emergy research. Ecol. Eng. 90, 285–293.
- Christie, M., Fazey, I., Cooper, R., Hyde, T., Kenter, J.O., 2012. An evaluation of monetary and non-monetary techniques for assessing the importance of biodiversity and ecosystem services to people in countries with developing economies. Ecol. Econ. 83, 67–78.
- Collins, A., Galli, A., Patrizi, N., Pulselli, F.M., 2018. Learning and teaching sustainability: the contribution of Ecological Footprint calculators. J. Clean. Prod. 174, 1000–1010.
- Coscieme, L., Pulselli, F.M., Marchettini, N., Sutton, P.C., Anderson, S., Sweeney, S., 2014. Emergy and ecosystem services: a national biogeographical assessment. Ecosyst. Serv. 7, 152–159.
- Costanza, R., D, R., Arge, de Groot, R., Farber, S., Grasso, M., 1997. The value of the world's ecosystem services and natural capital. Nature 6630, 253–260.
- Costanza, R., Daly, H.E., 1992. Natural capital and sustainable development. Conserv. Biol. 1, 37–46.
- Costanza, R., de Groot, R., Braat, L., Kubiszewski, I., Fioramonti, L., Sutton, P., Farber, S., Grasso, M., 2017. Twenty years of ecosystem services: how far have we come and how far do we still need to go? Ecosyst. Serv. 28, 1–16.
- Costanza, R., de Groot, R., Sutton, P., Van der Ploeg, S., Anderson, S.J., Kubiszewski, I., Farber, S., Turner, R.K., 2014. Changes in the global value of ecosystem services. Glob. Environ. Chang. 26, 152–158.
- Costanza, R., Farber, S., Liu, S., Troy, A., 2010. Valuing ecosystem services: theory, practice, and the need for a transdisciplinary synthesis. Ann. NY. Acad. Sci. 1, 54–78.
- Daily, G.C. (Ed.), 1997. Nature's Services: Societal Dependence on Natural Ecosystems. Island Press, Washingdon D.C.
- de Araujo Barbosa, C.C., Atkinson, P.M., Dearing, J.A., 2015. Remote sensing of ecosystem services: a systematic review. Ecol. Indicat. 52, 430–443.
- DesRoches, C.T., 2015. The World as a Garden: a Philosophical Analysis of Natural Capital in Economics. University of British Columbia.
- Dewulf, J., Benini, L., Mancini, L., Sala, S., Blengini, G.A., Ardente, F., Recchioni, M., Maes, J., Pant, R., Pennington, D., 2015. Rethinking the area of protection "natural resources" in life cycle assessment. Environ. Sci. Technol. 49, 5310–5317.
- Díaz, S., Demissew, S., Carabias, J., Joly, C., Lonsdale, M., Ash, N., Larigauderie, A., Adhikari, J.R., Arico, S., Báldi, A., 2015. The IPBES Conceptual Framework—connecting nature and people. Curr. Opin. Env. Sust. 14, 1–16.
- Ellegaard, O., 2018. The application of bibliometric analysis: disciplinary and user aspects. Scientometrics 1–22.
- Ellegaard, O., Wallin, J.A., 2015. The bibliometric analysis of scholarly production:

how great is the impact? Scientometrics 105, 1809-1831.

- Fahimnia, B., Sarkis, J., Davarzani, H., 2015. Green supply chain management: a review and bibliometric analysis. Int. J. Prod. Econ. 162, 101–114.
- Fang, K., Heijungs, R., 2015. Investigating the inventory and characterization aspects of footprinting methods: lessons for the classification and integration of footprints. J. Clean. Prod. 108, 1028–1036.
- Fang, K., Song, S., Heijungs, R., de Groot, S., Dong, L., Song, J., Wiloso, E.I., 2016. The footprint's fingerprint: on the classification of the footprint family. Curr. Opin. Env. Sust. 23, 54–62.
- Fang, K., Zhang, Q., Yu, H., Wang, Y., Dong, L., Shi, L., 2018. Sustainability of the use of natural capital in a city: measuring the size and depth of urban ecological and water footprints. Sci. Total Environ. 631, 476–484.
- Fisher, B., Turner, R.K., Morling, P., 2009. Defining and classifying ecosystem services for decision making. Ecol. Econ. 68, 643–653.
- Frazier, A.E., Renschler, C.S., Miles, S.B., 2013. Evaluating post-disaster ecosystem resilience using MODIS GPP data. Int. J. Appl. Earth Obs. 1, 43–52.
- Frélichová, J., Vačkář, D., Pártl, A., Loučková, B., Harmáčková, Z.V., Lorencová, E., 2014. Integrated assessment of ecosystem services in the Czech Republic. Ecosyst. Serv. 8, 110–117.
- Fu, B., Wang, Y.K., Xu, P., Yan, K., Li, M., 2014. Value of ecosystem hydropower service and its impact on the payment for ecosystem services. Sci. Total Environ. 472, 338–346.
- Fu, H., Ho, Y., 2013. Independent research of China in science citation index expanded during 1980–2011. J. Informetr. 7, 210–222.
- Galli, A., Giampietro, M., Goldfinger, S., Lazarus, E., Lin, D., Saltelli, A., Wackernagel, M., Müller, F., 2016. Questioning the ecological footprint. Ecol. Indicat. 69, 224–232.
- Galli, A., Halle, M., Grunewald, N., 2015. Physical limits to resource access and utilisation and their economic implications in Mediterranean economies. Environ. Sci. Policy 51, 125–136.
- Galli, A., Iha, K., Halle, M., El Bilali, H., Grunewald, N., Eaton, D., Capone, R., Debs, P., Bottalico, F., 2017. Mediterranean countries' food consumption and sourcing patterns: an Ecological Footprint viewpoint. Sci. Total Environ. 578, 383–391.
- Galos, K., Nieć, M., Saługa, P.W., Uberman, R., 2015. The basic problems of mineral resources valuation methodologies within the framework of System of Integrated Environmental and Economic Accounts. Gospod. Surowcami Miner. 31, 5–20.
- Geissdoerfer, M., Savaget, P., Bocken, N.M., Hultink, E.J., 2017. The Circular Economy–A new sustainability paradigm? J. Clean. Prod. 143, 757–768.
- Goldstein, J.H., Caldarone, G., Duarte, T.K., Ennaanay, D., Hannahs, N., Mendoza, G., Polasky, S., Wolny, S., Daily, G.C., 2012. Integrating ecosystem-service tradeoffs into land-use decisions. Proc. Natl. Acad. Sci. U.S.A. 109, 7565–7570.
- Gómez-Baggethun, E., de Groot, R., Lomas, P.L., Montes, C., 2010. The history of ecosystem services in economic theory and practice: from early notions to markets and payment schemes. Ecol. Econ. 69, 1209–1218.
- Grafius, D.R., Corstanje, R., Warren, P.H., Evans, K.L., Hancock, S., Harris, J.A., 2016. The impact of land use/land cover scale on modelling urban ecosystem services. Landsc. Ecol. 31, 1509–1522.
- Guerry, A.D., Polasky, S., Lubchenco, J., Chaplin-Kramer, R., Daily, G.C., Griffin, R., Ruckelshaus, M., Bateman, I.J., Duraiappah, A., Elmqvist, T., 2015. Natural capital and ecosystem services informing decisions: from promise to practice. Proc. Natl. Acad. Sci. U.S.A. 112, 7348–7355.
- Haines-Young, R., Potschin, M., 2012. Common International Classification of Ecosystem Services (CICES, Version 4.1). European Environmental Agency, Nottingham.
- Hamel, P., Chaplin-Kramer, R., Sim, S., Mueller, C., 2015. A new approach to modeling the sediment retention service (InVEST 3.0): case study of the Cape Fear catchment, North Carolina, USA. Sci. Total Environ. 524, 166–177.
- Häyhä, T., Franzese, P.P., 2014. Ecosystem services assessment: a review under an ecological-economic and systems perspective. Ecol. Model. 289, 124–132.
- Hein, L., Bagstad, K., Edens, B., Obst, C., de Jong, R., Lesschen, J.P., 2016. Defining ecosystem assets for natural capital accounting. PLoS One 11 e164460.
- Howe, C., Suich, H., Vira, B., Mace, G.M., 2014. Creating win-wins from trade-offs? Ecosystem services for human well-being: a meta-analysis of ecosystem service trade-offs and synergies in the real world. Glob. Environ. Chang. 28, 263–275.
- Hoyer, R., Chang, H., 2014. Assessment of freshwater ecosystem services in the Tualatin and Yamhill basins under climate change and urbanization. Appl. Geogr. 53, 402–416.
- ISO, 2006. Environmental Management: Life Cycle Assessment: Principles and Framework. International Organization for Standardization, Geneva.
- Jørgensen, S.E., 2010. Ecosystem services, sustainability and thermodynamic indicators. Ecol. Complex. 7, 311–313.
- Kenter, J.O., O'Brien, L., Hockley, N., Ravenscroft, N., Fazey, I., Irvine, K.N., Reed, M.S., Christie, M., Brady, E., Bryce, R., 2015. What are shared and social values of ecosystems? Ecol. Econ. 111, 86–99.
- Koellner, T., De Baan, L., Beck, T., Brandão, M., Civit, B., Margni, M., I Canals, L.M., Saad, R., De Souza, D.M., Müller-Wenk, R., 2013. UNEP-SETAC guideline on global land use impact assessment on biodiversity and ecosystem services in LCA. Int. J. Life Cycle Assess. 18, 1188–1202.
- La Notte, A., Maes, J., Dalmazzone, S., Crossman, N.D., Grizzetti, B., Bidoglio, G., 2017. Physical and monetary ecosystem service accounts for Europe: a case study for in-stream nitrogen retention. Ecosyst. Serv. s 23, 18–29.
- Lai, T., Salminen, J., Jäppinen, J., Koljonen, S., Mononen, L., Nieminen, E., Vihervaara, P., Oinonen, S., 2018. Bridging the gap between ecosystem service

indicators and ecosystem accounting in Finland. Ecol. Model. 377, 51-65.

- Landers, D.H., Nahlik, A.M., 2013. Final Ecosystem Goods and Services Classification System (FEGS-CS). United States Environmental Protection Agency, Washington DC.
- Lin, D., Hanscom, L., Murthy, A., Galli, A., Evans, M., Neill, E., Mancini, M., Martindill, J., Medouar, F., Huang, S., 2018. Ecological footprint accounting for countries: updates and results of the national footprint accounts, 2012–2018. Resources 7, 58.
- Liu, X., Ziv, G., Bakshi, B.R., 2018a. Ecosystem services in life cycle assessment-Part 1: a computational framework. J. Clean. Prod. 197, 314–322.
- Liu, X., Ziv, G., Bakshi, B.R., 2018b. Ecosystem services in life cycle assessment-Part 2: adaptations to regional and serviceshed information. J. Clean. Prod. 197, 772–780.
- Liu, Y., Fu, B., Zhao, W., Wang, S., 2018. Ecological asset accounting and ecosystem services evaluation: concept intersection and key research priorities. Acta Ecol. Sin. 38, 8267–8276.
- Lu, H., Fu, F., Li, H., Campbell, D.E., Ren, H., 2015. Eco-exergy and emergy based selforganization of three forest plantations in lower subtropical China. Sci. Rep.-UK 5, 15047.
- Lu, N., Fu, B., Jin, T., Chang, R., 2014. Trade-off analyses of multiple ecosystem services by plantations along a precipitation gradient across Loess Plateau landscapes. Landsc. Ecol. 29, 1697–1708.
- Mancini, M.S., Galli, A., Coscieme, L., Niccolucci, V., Lin, D., Pulselli, F.M., Bastianoni, S., Marchettini, N., 2018. Exploring ecosystem services assessment through Ecological Footprint accounting. Ecosyst. Serv. 30, 228–235.
- Mancini, M.S., Galli, A., Niccolucci, V., Lin, D., Hanscom, L., Wackernagel, M., Bastianoni, S., Marchettini, N., 2017. Stocks and flows of natural capital: implications for ecological footprint. Ecol. Indicat. 77, 123–128.
- Mao, G., Shi, T., Zhang, S., Crittenden, J., Guo, S., Du, H., 2018. Bibliometric analysis of insights into soil remediation. J. Soils Sediments 18, 2520–2534.
- Mao, G., Zou, H., Chen, G., Du, H., Zuo, J., 2015. Past, current and future of biomass energy research: a bibliometric analysis. Renew. Sustain. Energy Rev. 52, 1823–1833.
- Maseyk, F.J., Mackay, A.D., Possingham, H.P., Dominati, E.J., Buckley, Y.M., 2017. Managing natural capital stocks for the provision of ecosystem services. Conserv. Lett. 10, 211–220.
- McLinden, D., 2013. Concept maps as network data: analysis of a concept map using the methods of social network analysis. Eval. Program Plann. 36, 40–48.
- MEA, 2005. Millennium Ecosystem Assessment: Ecosystems and Human Well-Being. World Health Organization, Geneva.
- Mellino, S., Buonocore, E., Ulgiati, S., 2015. The worth of land use: a GIS-emergy evaluation of natural and human-made capital. Sci. Total Environ. 506, 137–148.
- Missemer, A., 2018. Natural capital as an economic concept, history and contemporary issues. Ecol. Econ. 143, 90–96.
- Mo, Z., Fu, H., Ho, Y., 2018. Global development and trend of wind tunnel research from 1991 to 2014: a bibliometric analysis. Environ. Sci. Pollut. Res. 25, 30257–30270.
- Mouchet, M.A., Lamarque, P., Martín-López, B., Crouzat, E., Gos, P., Byczek, C., Lavorel, S., 2014. An interdisciplinary methodological guide for quantifying associations between ecosystem services. Glob. Environ. Chang. 28, 298–308.
- Müller-Wenk, R., Brandão, M., 2010. Climatic impact of land use in LCA—carbon transfers between vegetation/soil and air. Int. J. Life Cycle Assess. 15, 172–182.
- Niccolucci, V., Bastianoni, S., Tiezzi, E., Wackernagel, M., Marchettini, N., 2009. How deep is the footprint? A 3D representation. Ecol. Model. 220, 2819–2823.
- Obst, C., Hein, L., Edens, B., 2016. National accounting and the valuation of ecosystem assets and their services. Environ. Resour. Econ. 64, 1–23.
- Obst, C., Vardon, M., 2014. Recording environmental assets in the national accounts. Oxf. Rev. Econ. Pol. 30, 126–144.
- Odum, H.T., Brown, M.T., Williams, S.B., 2000. Handbook of Emergy Evaluation. University of Florida, Florida.
- Ouyang, Z., Zheng, H., Xiao, Y., Polasky, S., Liu, J., Xu, W., Wang, Q., Zhang, L., Xiao, Y., Rao, E., 2016. Improvements in ecosystem services from investments in natural capital. Science 352, 1455–1459.
- Pascual, U., Balvanera, P., Díaz, S., Pataki, G., Roth, E., Stenseke, M., Watson, R.T., Dessane, E.B., Islar, M., Kelemen, E., 2017. Valuing nature's contributions to people: the IPBES approach. Curr. Opin. Env. Sust. 26, 7–16.
- Pascual, U., Muradian, R., Brander, L., Gómez-Baggethun, E., Martín-López, B., Verma, M., Armsworth, P., Christie, M., Cornelissen, H., Eppink, F., 2010. The Economics of Valuing Ecosystem Services and Biodiversity. TEEB–Ecological and Economic Foundation, Washington DC.
- Pedro-Monzonís, M., Solera, A., Ferrer, J., Andreu, J., Estrela, T., 2016. Water accounting for stressed river basins based on water resources management models. Sci. Total Environ. 565, 181–190.
- Persson, O., Danell, R., Schneider, J.W., 2009. How to use Bibexcel for various types of bibliometric analysis. Celebrating scholarly communication studies: A Fest-schrift for Olle Persson at his 60th Birthday 5, 9–24.
- Pizzirani, S., McLaren, S.J., Seadon, J.K., 2014. Is there a place for culture in life cycle sustainability assessment? Int. J. Life Cycle Assess. 19, 1316–1330.
- Polasky, S., Bryant, B., Hawthorne, P., Johnson, J., Keeler, B., Pennington, D., 2015. Inclusive wealth as a metric of sustainable development. Annu. Rev. Environ. Resour. 40, 445–466.
- Quinteiro, P., Dias, A.C., Ridoutt, B.G., Arroja, L., 2014. A framework for modelling the transport and deposition of eroded particles towards water systems in a life cycle inventory. Int. J. Life Cycle Assess. 19, 1200–1213.
- Raugei, M., Rugani, B., Benetto, E., Ingwersen, W.W., 2014. Integrating emergy into

LCA: potential added value and lingering obstacles. Ecol. Model. 271, 4–9.

- Remme, R.P., Edens, B., Schröter, M., Hein, L., 2015. Monetary accounting of ecosystem services: a test case for Limburg province, The Netherlands. Ecol. Econ. 112, 116–128.
- Reza, B., Sadiq, R., Hewage, K., 2014. Emergy-based life cycle assessment (Em-LCA) for sustainability appraisal of infrastructure systems: a case study on paved roads. Clean Technol. Envir. 16, 251–266.
- Richardson, L., Loomis, J., Kroeger, T., Casey F, 2015. The role of benefit transfer in ecosystem service valuation. Ecol. Econ. 115, 51–58.
- Robinson, D.A., Fraser, I., Dominati, E.J., Davíðsdóttir, B., Jónsson, J., Jones, L., Jones, S.B., Tuller, M., Lebron, I., Bristow, K.L., 2014. On the value of soil resources in the context of natural capital and ecosystem service delivery. Soil Sci. Soc. Am. J. 78, 685–700.
- Rugani, B., Roviani, D., Hild, P., Schmitt, B., Benetto, E., 2014. Ecological deficit and use of natural capital in Luxembourg from 1995 to 2009. Sci. Total Environ. 468, 292–301.
- Saad, R., Koellner, T., Margni, M., 2013. Land use impacts on freshwater regulation, erosion regulation, and water purification: a spatial approach for a global scale level. Int. J. Life Cycle Assess. 18, 1253–1264.
- Schaefer, M., Goldman, E., Bartuska, A.M., Sutton-Grier, A., Lubchenco, J., 2015. Nature as capital: advancing and incorporating ecosystem services in United States federal policies and programs. Proc. Natl. Acad. Sci. U.S.A. 7383–7389.
- Schaubroeck, T., Rugani, B., 2017. A revision of what life cycle sustainability assessment should entail: towards modeling the net impact on human wellbeing. J. Ind. Ecol. 21, 1464–1477.
- Schröter, M., van der Zanden, E.H., van Oudenhoven, A.P., Remme, R.P., Serna Chavez, H.M., De Groot, R.S., Opdam, P., 2014. Ecosystem services as a contested concept: a synthesis of critique and counter-arguments. Conserv. Lett. 7, 514–523.
- Schultz, L., Folke, C., Österblom, H., Olsson, P., 2015. Adaptive governance, ecosystem management, and natural capital. Proc. Natl. Acad. Sci. U.S.A. 7369–7374.
- Shepherd, E., Milner Gulland, E.J., Knight, A.T., Ling, M.A., Darrah, S., van Soesbergen, A., Burgess, N.D., 2016. Status and trends in global ecosystem services and natural capital: assessing progress toward Aichi biodiversity target 14. Conserv. Lett. 9, 429–437.
- Sherrouse, B.C., Semmens, D.J., Clement, J.M., 2014. An application of social values for ecosystem services (SolVES) to three national forests in Colorado and Wyoming. Ecol. Indicat. 36, 68–79.
- Shoyama, K., Kamiyama, C., Morimoto, J., Ooba, M., Okuro, T., 2017. A review of modeling approaches for ecosystem services assessment in the Asian region. Ecosyst. Serv. 26, 316–328.
- Smith, A.C., Harrison, P.A., Soba, M.P., Archaux, F., Blicharska, M., Egoh, B.N., Erős, T., Domenech, N.F., György, Á.I., Haines-Young, R., 2017. How natural capital delivers ecosystem services: a typology derived from a systematic review. Ecosyst. Serv. 26, 111–126.
- Stadler, K., Wood, R., Bulavskaya, T., Södersten, C.J., Simas, M., Schmidt, S., Usubiaga, A., Acosta Fernández, J., Kuenen, J., Bruckner, M., 2018. EXIOBASE 3: developing a time series of detailed environmentally extended multi-regional input-output tables. J. Ind. Ecol. 22, 502–515.
- Tallis, H., Lubchenco, J., 2014. Working together: a call for inclusive conservation. Nature News 515, 27.
- TEEB, 2010. The Economics of Ecosystems and Biodiversity: Ecological and Economic Foundations. Earchscan, London & Washiongton, DC.
- Ukidwe, N.U., Bakshi, B.R., 2007. Industrial and ecological cumulative exergy consumption of the United States via the 1997 input–output benchmark model. Energy 32, 1560–1592.
- UN, 2014a. System of Environmental-Economic Accounting 2012: Central Framework. United Nations Publications, New York.
- UN, 2014b. System of Environmental–Economic Accounting 2012—experimental Ecosystem Accounting. United Nations Publications, New York.
- Valero, A., Valero, A., Stanek, W., 2018. Assessing the exergy degradation of the natural capital: from Szargut's updated reference environment to the new

thermoecological-cost methodology. Energy 163, 1140-1149.

- Van Berkel, D.B., Verburg, P.H., 2014. Spatial quantification and valuation of cultural ecosystem services in an agricultural landscape. Ecol. Indicat. 37, 163–174.
- Vardon, M., Burnett, P., Dovers, S., 2016. The accounting push and the policy pull: balancing environment and economic decisions. Ecol. Econ. 124, 145–152.
- Vassallo, P., Paoli, C., Buonocore, E., Franzese, P.P., Russo, G.F., Povero, P., 2017. Assessing the value of natural capital in marine protected areas: a biophysical and trophodynamic environmental accounting model. Ecol. Model. 355, 12–17.
- Venetoulis, J., Talberth, J., 2008. Refining the ecological footprint. Environ. Dev. Sustain. 10, 441–469.
- Villa, F., Bagstad, K.J., Voigt, B., Johnson, G.W., Portela, R., Honzák, M., Batker, D., 2014. A methodology for adaptable and robust ecosystem services assessment. PLoS One 9, e91001.
- Wackernagel, M., Galli, A., Hanscom, L., Lin, D., Mailhes, L., Drummond, T., 2018a. Ecological footprint accounts: criticisms and applications. In: Bell, S., Morse, S. (Eds.), Routledge Handbook of Sustainability Indicators. Routledge International Handbooks, pp. 521–539.
- Wackernagel, M., Galli, A., Hanscom, L., Lin, D., Mailhes, L., Drummond, T., 2018b. Ecological footprint accounts: principles. In: Bell, S., Morse, S. (Eds.), Routledge Handbook of Sustainability Indicators. Routledge International Handbooks, pp. 244–264.
- Wackernagel, M., Rees, W.E., 1997. Perceptual and structural barriers to investing in natural capital: economics from an ecological footprint perspective. Ecol. Econ. 20, 3–24.
- Wallace, K.J., 2007. Classification of ecosystem services: problems and solutions. Biol. Conserv. 139, 235–246.
- Wang, C., Li, X., Yu, H., Wang, Y., 2019. Tracing the spatial variation and value change of ecosystem services in Yellow River Delta, China. Ecol. Indicat. 96, 270–277.
- Wang, C., Wang, Y., Geng, Y., Wang, R., Zhang, J., 2016. Measuring regional sustainability with an integrated social-economic-natural approach: a case study of the Yellow River Delta region of China. J. Clean. Prod. 114, 189–198.
- Wang, L., Zhao, L., Mao, G., Zuo, J., Du, H., 2017. Way to accomplish low carbon development transformation: a bibliometric analysis during 1995–2014. Renew. Sustain. Energy Rev. 68, 57–69.
- Wang, R., Li, F., Hu, D., Larry Li, B., 2011. Understanding eco-complexity: socialeconomic-natural complex ecosystem approach. Ecol. Complex. 8, 15–29.
- Watanabe, M.D., Ortega, E., 2014. Dynamic emergy accounting of water and carbon ecosystem services: a model to simulate the impacts of land-use change. Ecol. Model. 271, 113–131.
- Wei, Y., Wang, J., Chen, T., Yu, B., Liao, H., 2018. Frontiers of low-carbon technologies: results from bibliographic coupling with sliding window. J. Clean. Prod. 190, 422–431.
- Wiedmann, T.O., Schandl, H., Lenzen, M., Moran, D., Suh, S., West, J., Kanemoto, K., 2015. The material footprint of nations. Proc. Natl. Acad. Sci. U.S.A. 112, 6271–6276.
- Wood, R., Stadler, K., Bulavskaya, T., Lutter, S., Giljum, S., de Koning, A., Kuenen, J., Schütz, H., Acosta-Fernández, J., Usubiaga, A., 2015. Global sustainability accounting—developing EXIOBASE for multi-regional footprint analysis. Sustainability-Basel 7, 138–163.
- Yang, Y., Hu, D., 2018. Natural capital utilization based on a three-dimensional ecological footprint model: a case study in northern Shaanxi, China. Ecol. Indicat. 87, 178–188.
- Zhang, J., Gurkan, Z., Jørgensen, S.E., 2010a. Application of eco-exergy for assessment of ecosystem health and development of structurally dynamic models. Ecol. Model. 221, 693–702.
- Zhang, Y., Huang, K., Yu, Y., Yang, B., 2017. Mapping of water footprint research: a bibliometric analysis during 2006–2015. J. Clean. Prod. 149, 70–79.
- Zhang, Y.I., Singh, S., Bakshi, B.R., 2010b. Accounting for ecosystem services in life cycle assessment, Part I: a critical review. Environ. Sci. Technol. 44, 2232–2242.
- Zhong, S., Geng, Y., Liu, W., Gao, C., Chen, W., 2016. A bibliometric review on natural resource accounting during 1995–2014. J. Clean. Prod. 139, 122–132.